Author's response to reviews

Title: A 10-year time-series analysis of respiratory and cardiovascular morbidity in Nicosia, Cyprus: The effect of short-term changes in air pollution and dust storms.

Authors:

Nicos Middleton (nmitlett@hsph.harvard.edu)
Panayiotis Yiallouros (pyiallouros@cyprusinstitute.org)
Savvas Kleanthous (skleanthous@dl.mlsi.gov.cy)
Ourania Kolokotroni (okolokotroni@cyprusinstitute.org)
Joel Schwartz (jschwrtz@hsph.harvard.edu)
Douglas W Dockery (ddockery@hsph.harvard.edu)
Phil Demokritou (pdemokri@hsph.harvard.edu)
Petros Koutrakis (petros@hsph.harvard.edu)

Version: 3 Date: 7 July 2008

Author's response to reviews: see over
7 July 2008

The Editor in Chief, Environmental Health

Dear Editor,

MS: 2815064211799503, Research
A 10-year time series analysis of respiratory and cardiovascular morbidity in Nicosia, Cyprus: the effects of short-term changes in air pollution and dust storms.

We have now made all the changes listed in your email dated 25 June 2008 and we are pleased to resubmit a revised version of the above paper. As requested, changes include double spacing the abstract, removing titles from the figures and re-formatting the references. Regarding the latter, I would like to bring to your attention that while we have removed issue numbers from the references in the manuscript as instructed, these seem to still appear in the extracted references of the uploaded version. We haven’t been able to correct this. I expect the publisher will let us know of any additional items that need attention. We look forward to receiving your response.

Yours sincerely,

Nicos Middleton, PhD